Author’s response to reviews

Title: Cenal macular thickness in patients with type 2 diabetes mellitus without clinical retinopathy

Authors:

Mehmet Demir (drmehmetfe@hotmail.com)
Ersin Oba (ersinoba@yahoo.com)
Burcu Dirim (burcu_dirim@hotmail.com)
Erhan Ozdal (eozdal2002@gmail.com)
Efe Can (efecan06@hotmail.com)

Version: 6 Date: 5 March 2013

Author’s response to reviews: see over
“Dear Dr. Demir,

Thank you so much for the revised manuscript. Our editors have been happy with your manuscript. We still need more essential revisions in order for your manuscript to be accepted”:

1) Please include the title page in the main manuscript.

   Answer:

   Title page attached in the main text.

2) Please expand the background section of the abstract to include some context information.

   Answer:

   Some context information added into the background section of the abstract and it was expanded.

We would be grateful if you could address the comments in a revised manuscript and provide a cover letter giving a point-by-point response to the concerns.

Thank you for your help. We revised manuscript according to comments.

Best regards